Author’s response to reviews

Title: Specialty cardiac nurses' work satisfaction is influenced by the type of coronary care unit: A mixed methods study

Authors:

Judy Currey (judy.currey@deakin.edu.au)
Stephanie Sprogis (s.sprogis@deakin.edu.au)
Liliana Orellana (l.orellana@deakin.edu.au)
Anusha Chander (anusha.chander@outlook.com)
Sharon Meagher (s.meagher@deakin.edu.au)
Rhoda Kennedy (rhoda.kennedy@deakin.edu.au)
Andrea Driscoll (andrea.driscoll@deakin.edu.au)

Version: 1 Date: 10 Feb 2019

Author’s response to reviews:

Please refer to the cover letter submitted with this manuscript revision. A point by point response to the comments has been included within the letter as requested from the journal via email.